Author's response to reviews

Title: Molecular characterization of Histoplasma capsulatum isolated from an outbreak in treasure hunters

Authors:

Bertha Muñoz (bermh_fungi2002@yahoo.com.mx)
Maria A Martínez (angeles.12mar@hotmail.com)
Gabriel Palma (gabpal52@yahoo.com.mx)
Amado Ramírez (arh_2@hotmail.com)
Maria G Frías (mariaguadalupefrias@hotmail.com)
Maria R Reyes (re moa@servidor.unam.mx)
Maria L Taylor (emello@servidor.unam.mx)
Anjarath L Higuera (ahigeraiglesias@yahoo.com.mx)
Alexander Corcho (aleksenrique@yahoo.com)
Maria E Manjarrez (e_manjarrez@yahoo.com)

Version: 6 Date: 31 August 2010

Author's response to reviews: see over
The BMC Series Editorial Production Team

I send the covering letter with a point-by-point description of the changes made.

Major revisions.

1. References: reference 25 contains the details for multiple sources, please separate the details into separate reference details, and update the text accordingly, so that each source is cited correctly.

Since the reference 25 was corrected. The 26 had run

Minor revisions

1. Highlighting/tracking: Please remove all highlighting and tracking from the manuscript.

The text is clean of all highlighting and tracking

2. Figures: It is important for the final layout of the manuscript that the figures are cropped as closely as possible to minimise white space around the image. Our online figure guide contains full details for preparing files for submission and can be viewed here: http://www.biomedcentral.com/info/ifora/figures

The figures were cropped as closely as possible and I Consult the page and see the guide and I preparing files for submission.

3. Typography: Please take this opportunity to check your manuscript for any typographical errors and to make any final corrections or revisions. This is the final proofing stage for your manuscript, and you will not be able to make any changes after acceptance.

I check the manuscript for errors

Thanks
Ma. Eugenia Manjarrez Zavala